Author’s response to reviews

Title: Cold exposure induced oxidative stress and apoptosis in the myocardium by inhibiting the Nrf2-Keap1 signaling pathway

Authors:
Peifang Cong (karyscorpio@163.com)
Yunen Liu (lye9901@163.com)
Nannan Liu (renwuxiaohao002@163.com)
Yubiao Zhang (13654907765@qq.com)
Changci Tong (815176868@qq.com)
Lin Shi (460087525@qq.com)
Xuelei Liu (313295878@qq.com)
Xiuyun Shi (1228689471@qq.com)
Ying Liu (752409195@qq.com)
Zhou Tong (123888913@qq.com)
Mingxiao Hou (houmingxiao188@163.com)

Version: 1 Date: 28 Aug 2017

Author’s response to reviews:

Dear editors,

The article has been revised follow editor's comments:

1. Figures were removed from the text.
2. page numbering was added.
3. font size revised higher.
4. references has changed to properly format
5. Authors' Information has been added in the Declarations.
And other parts of the article were confirmed again following the editorial policies.